Author's response to reviews

Title: Implementation of a multicomponent intervention to prevent physical restraints in nursing home residents (IMPRINT): study protocol for a cluster-randomised controlled trial

Authors:

Jens Abraham (jens.abraham@medizin.uni-halle.de)
Ralph Mührer (ralph.moehler@uni-wh.de)
Adrienne Henkel (adrienne.henkel@uksh.de)
Ramona Kupfer (ramona.kupfer@uksh.de)
Andrea Icks (andrea.icks@uni-duesseldorf.de)
Charalabos-Markos Dintios (chmdintsios@aol.com)
Burkhard Haastert (haastert@medistatistica.de)
Gabriele Meyer (gabriele.meyer@medizin.uni-halle.de)
Sascha Köpke (sascha.koepke@uksh.de)

Version: 2
Date: 16 June 2015

Author's response to reviews: see over
Dear Sir or Madam

Thank you for your comments to our submitted protocol. Hereinafter you find our responds to your comments.

1. the study should adhere to the CONSORT guidelines and the flowchart should be formatted accordingly

Our study adheres to the CONSORT guidelines. The CONSORT 2010 checklist of information to include when reporting a cluster randomised trial has been completed accordingly and submitted as additional file. We have made a few changes to our manuscript: the Name of trial registry (page 4) and the Person who will generate the randomisation list (page 10) have been added.

The flowchart adheres to CONSORT guidelines.

2. Trial status does not need to be included in the main text of the manuscript

Trial status has been excluded from the main text of the manuscript.

3. please clarify the contribution of Charalabos-Markos Dintios to the study according to our editorial policies for authorship

The author contribution of Charalabos-Markos Dintios has been added accordingly (page 19).
We hope that we have processed your comments satisfactorily and look forward for your response.

With best regards

Gabriele Meyer